Author's response to reviews

Title: Atopic dermatitis, cutaneous steroids and cataracts in children: a case series.

Authors:

Andrew Tatham (andrewtatham@yahoo.co.uk)

Version: 2 Date: 18 January 2008

Author's response to reviews:

This manuscript has been renamed 'a case series' as it contains two cases. I have also included confirmation that informed consent was obtained for these patients.
This changes were as suggested by the editor.

Yours faithfully

Andrew Tatham